Author's response to reviews

Title: Missing mercury from mercury cell chlor-alkali plants may be found in food products: an environmental investigation of added sugar in foods

Authors:

Renee Dufault (rdufault@phaccountability.org)
Blaise LeBlanc (blaise_ll@hotmail.com)
Roseanne Schnoll (rschnoll@brooklyn.cuny.edu)
Charles Cornett (cornettc@uwplatt.edu)
Laura Schweitzer (schweitl@uwplatt.edu)
Lyn Patrick (lpatrick@frontier.net)
Jane Hightower (jhightowermd@aol.com)
David Wallinga (dwallinga@iatp.org)
Walter Lukiw (wlukiw@lsuhsc.edu)

Version: 2 Date: 10 September 2008

Author's response to reviews:

Please do NOT send this manuscript to anyone affiliated with the US Food and Drug Administration. Please DO send it to the following reviewers who have agreed to review it:

Dr. Jim Stevenson at the University of South Hampton at jsteven@soton.ac.uk
Dr. Raymond Palmer at the University of Texas at palmerr@uthscsa.edu
You can also send it to Chunying Chen at the Chinese Academy of Sciences at chunying.chen@mail.ihep.ac.cn

Thank you.

Kind Regards,

Renee Dufault